Answer to reviewer

I would like to express our sincere gratitude to the reviewer and the editorial team of the World Journal of Gastroenterology for the valuable feedback and constructive comments on our editorial titled “The Invisible Threat: Understanding Microplastics and Nanoplastics in Our Environment” by Chiang et al., published earlier this year.

We have carefully considered each comment and suggestion provided by the reviewer and have made the necessary revisions to our manuscript accordingly. Below, I outline the key changes and how we have addressed the concerns raised:

Titles: We have incorporated both a title and a short title into our submission as requested.

References: We have updated our reference list to ensure the most relevant and current literature is cited. Additionally, we have meticulously checked each reference for completeness and adherence to the journal’s style guide.

English and Typing Errors: The manuscript has been thoroughly reviewed and proofread to correct English language and typing errors. This task was entrusted to a colleague born and educated in Great Britain, ensuring the highest standard of language quality.
We believe these revisions have significantly improved the quality and clarity of our manuscript. We are resubmitting our manuscript for your consideration and eagerly look forward to the possibility of publication in the World Journal of Gastroenterology.

Thank you once again for your attention and support. Please do not hesitate to contact me if you require further information or if there are any additional requirements we need to fulfill.

Warm regards,

Jean Demarquoy Ph.D.